**Author’s response to reviews**

**Title:** Atraumatic patellar prosthesis dislocation with patellar tendon injury following a total knee arthroplasty: a case report

**Authors:**

Vinay Kumar Singh MBBS, MRCS, MCh(Orth) (orthopaediks@yahoo.co.uk)
Pankaj Kumar Singh MBBS, MS, MRCS (we.publish@googlemail.com)
Yashwant Singh MBBS, MRCS (dryashwant2006@yahoo.com)
Alka Singh MBBS (alka.singh@bht.nhs.net)
Sadaf Javed MBBS (sadafjaved@doctors.org.uk)
Murad Abdunabi MBBS, FRCS (abdunabi@web.de)

**Version:** 2  **Date:** 4 December 2009

**Author’s response to reviews:**

1. Ethnicity of the patients has been added in case report.
2. Figures are uploaded as per requirement.